Author's response to reviews

Title: A perspective on SIDS pathogenesis. The hypotheses: plausibility and evidence

Authors:

Paul N Goldwater (paul.goldwater@health.sa.gov.au)

Version: 8 Date: 21 March 2011

Author's response to reviews:

Dear Dr Lee,

I have added a couple of brief points (highlighted in blue on page 15) which I think are valuable additions. The addition reads... “The role of testosterone also deserves close examination because of the hormone’s profound effect on immune function145 and most notably the hormone’s peak observed in male infants between the age of 2 and 4 months which elegantly coincides with the SIDS age peak. 146”

I would hope this small addition would not delay publication

Kind regards,

Paul Goldwater